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Brief summary:
* E944 is in it’s third week of running since the repair of the 25M beam stop.

* The recovery has been very rapid: We're approaching
the same POT levels that we saw before the hiatus.

» We can attribute this to the remarkable performance of the Booster Accelerator,
and the hard work of the people in the Proton Source Department.

* The beam uptime this past week was 87%; a bit low but still delivered
good integrated beam intensity.

*Major down time can be attributed to:
4/17 — Ml problems ~4 hrs.
4/18 — Booster Notcher ~4hrs.
4/19 — Linac RF ~2 hrs.
4/20 -- Linac RF2 repair ~8 hrs.

* Total Protons for the week are — 4.25X1018
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POT per Horn Pulse
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Goals for the remainder of the running period:

{ Collect0.6t00.9 X10” PoTin v mode.
Totals to date:

~ 0.2 X10”" PoT with no absorber Interference;
~0.7 x10™" PoT with absorber #10 down;
~ 0.6 x10™ PoT with absorber #10 and #7 down.

~0.062 x10” PoT since restarting the run .
¢ There are 105 days remaining before the start of the
summer shutdown on August 8. To reach our goal

welll need to record 2. 7t0 4. 710" PoT per week.
( In the latest week we received 4 . 25X 1018 PoT.

b

¢ Continue v analysis;

0 . . . .
CCQE; m production; and Neutral Current interaction x-sections .



